# School-based relationship and sexuality education intervention engaging adolescent boys for the reductions of teenage pregnancy: the JACK cluster RCT

Maria Lohan,<sup>1\*</sup> Kathryn Gillespie,<sup>1</sup> Áine Aventin,<sup>1</sup>
Aisling Gough,<sup>1</sup> Emily Warren,<sup>2</sup> Ruth Lewis,<sup>3</sup>
Kelly Buckley,<sup>4</sup> Theresa McShane,<sup>1</sup>
Aoibheann Brennan-Wilson,<sup>1</sup> Susan Lagdon,<sup>1,5</sup>
Linda Adara,<sup>4</sup> Lisa McDaid,<sup>3,6</sup> Rebecca French,<sup>2</sup>
Honor Young,<sup>4</sup> Clíona McDowell,<sup>7</sup> Danielle Logan,<sup>7</sup>
Sorcha Toase,<sup>7</sup> Rachael M Hunter,<sup>8</sup> Andrea Gabrio,<sup>9</sup>
Mike Clarke,<sup>7,10</sup> Liam O'Hare,<sup>11</sup> Chris Bonell,<sup>2</sup>
Julia V Bailey<sup>12</sup> and James White<sup>4,13</sup>

<sup>&</sup>lt;sup>1</sup>School of Nursing and Midwifery, Queen's University Belfast, Belfast, UK

<sup>&</sup>lt;sup>2</sup>Department of Public Health, Environments and Society, London School of Hygiene & Tropical Medicine, London, UK

<sup>&</sup>lt;sup>3</sup>Medical Research Council/Chief Scientist Office Social and Public Health Sciences Unit, University of Glasgow, Glasgow, UK

<sup>&</sup>lt;sup>4</sup>Centre for Development, Evaluation, Complexity and Implementation in Public Health Improvement, Cardiff University, Cardiff, UK

<sup>&</sup>lt;sup>5</sup>Ulster University, Belfast, UK

<sup>&</sup>lt;sup>6</sup>Institute for Social Science Research, University of Queensland, Brisbane, QLD, Australia

<sup>&</sup>lt;sup>7</sup>Northern Ireland Clinical Trials Unit, Belfast, UK

<sup>&</sup>lt;sup>8</sup>Health Economics Analysis and Research Methods Team, University College London, London, UK

<sup>&</sup>lt;sup>9</sup>Care and Public Health Research Institute (CAPHRI) School for Public Health and Primary Care, Maastricht University, Maastricht, the Netherlands

<sup>&</sup>lt;sup>10</sup>School of Medicine, Dentistry and Biomedical Sciences, Queen's University Belfast and Northern Ireland Clinical Trials Unit, Belfast, UK

<sup>&</sup>lt;sup>11</sup>School of Social Sciences, Education and Social Work, Queen's University Belfast, Belfast, UK

<sup>&</sup>lt;sup>12</sup>E-Health Unit, University College London, London, UK

<sup>&</sup>lt;sup>13</sup>Centre for Trials Research, Cardiff University, Cardiff, UK

<sup>\*</sup>Corresponding author m.lohan@qub.ac.uk

### Disclosure of interests of authors

**Full disclosure of interests:** Completed ICMJE forms for all authors, including all related interests, are available in the toolkit on the NIHR Journals Library report publication page at https://doi.org/10.3310/YWXQ8757.

**Primary conflicts of interest:** Maria Lohan has received grants or contracts, paid to their institution, from the World Health Organization, Medical Research Council, Foreign and Commonwealth and Development Office, Nuffield Trust Foundation, National Institute for Health and Care Research (NIHR) Public Health Research (PHR) programme (NIHR PHR 12/153/26) and Medical Research Council Global Challenges Research Fund. Ruth Lewis was a co-principal investigator on CONUNDRUM (CONdom and CONtraception UNderstandings: Researching Uptake and Motivations), which was funded by NHS Greater Glasgow and Clyde, NHS Lanarkshire and NHS Lothian, in partnership with the Scottish Government (2019–20). Kelly Buckley was a co-investigator on Family Recovery after Domestic Abuse (FReDA): a feasibility trial and nested process evaluation of a group based psychoeducational intervention for children exposed to domestic violence and abuse, which was funded by NIHR PHR programme grant number 127793 (payment made to Cardiff University). Kelly Buckley was also a co-investigator on the Health Pathfinder Evaluation, funded by Standing Together and Safelives (payment made to Cardiff University). Lisa McDaid has received grants or contracts from the Medical Research Council/Chief Scientist Office (MRC/CSO) Social and Public Health Sciences Unit, University of Glasgow (MC\_UU\_12017/11, SPHSU11; MC\_UU\_00022/3, SPHSU18). Rebecca French has received grants or contracts from NIHR Health and Social Care Delivery Research (NIHR129529) and Public Health England (The development and national pilot of a women's reproductive health tracking survey), and consulting fees (paid to London School of Hygiene & Tropical Medicine) from Pause (London, UK) for qualitative research with key stakeholders to explore conditional use of long-acting reversible contraceptives methods for women who have had children taken into care to continue on the Pause programme. Rebecca French is also chairperson of the Sexual & Reproductive Health Clinical Studies Group for the Faculty of Sexual and Reproductive Healthcare, Royal College of Obstetricians and Gynaecologists. Chris Bonell was a member of the PHR research funding board (2013–19).

Published September 2023 DOI: 10.3310/YWXQ8757

# Plain language summary

School-based relationship and sexuality education intervention engaging adolescent boys for the reductions of teenage pregnancy: the JACK cluster RCT

Public Health Research 2023; Vol. 11: No. 8

DOI: 10.3310/YWXQ8757

NIHR Journals Library www.journalslibrary.nihr.ac.uk

# **Plain language summary**

dolescent pregnancy is often thought to be an issue for young women alone, but it is important to engage young men to tackle the problem and find solutions. The If I Were Jack intervention was especially designed to engage with boys as well as girls aged 14 years and to promote positive masculinity and gender equality to prevent adolescent pregnancy and promote positive sexual health. It uses tailored interactive films and resources, made with the help of students and teachers, to make it relevant to each of the four UK nations. This relationship and sexuality education intervention encourages adolescents to avoid unprotected sex by delaying sexual activity until they feel ready and to use reliable contraception once sexually active. It also promotes knowledge, attitudes (such as beliefs about gender and masculinities), skills and intentions for safe and pleasurable relationships. In this trial, we compared students in 33 schools randomly allocated to deliver the intervention with students in 33 schools that continued with their usual relationship and sexuality education practices. Four schools withdrew, two because of COVID-19 school closures. This left a total of 6556 students who completed questionnaires at the start of the study and 12-14 months later. Responses from all these students showed that If I Were Jack had a positive impact on knowledge, attitudes and intentions required for safe and pleasurable relationships, but did not have a significant effect on overall avoidance of unprotected sex. This was because the intervention had no effect on delaying sexual activity. However, we found that the intervention was effective in increasing the use of reliable contraception as students became sexually active, as well as for those who already were sexually active prior to receiving the intervention. We also found that If I Were Jack was likely to provide value for money by reducing unintended pregnancies and improving sexual health.

## **Public Health Research**

ISSN 2050-4381 (Print)

ISSN 2050-439X (Online)

Public Health Research (PHR) was launched in 2013 and is indexed by Europe PMC, NCBI Bookshelf, DOAJ, INAHTA, Ulrichsweb™ (ProQuest LLC, Ann Arbor, MI, USA), and MEDLINE.

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full PHR archive is freely available to view online at www.journalslibrary.nihr.ac.uk/phr.

### Criteria for inclusion in the Public Health Research journal

Reports are published in *Public Health Research* (PHR) if (1) they have resulted from work for the PHR programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Reviews in *Public Health Research* are termed 'systematic' when the account of the search appraisal and synthesis methods (to minimise biases and random errors) would, in theory, permit the replication of the review by others.

### PHR programme

The Public Health Research (PHR) programme, part of the National Institute for Health and Care Research (NIHR), is the leading UK funder of public health research, evaluating public health interventions, providing new knowledge on the benefits, costs, acceptability and wider impacts of non-NHS interventions intended to improve the health of the public and reduce inequalities in health. The scope of the programme is multi-disciplinary and broad, covering a range of interventions that improve public health.

For more information about the PHR programme please visit the website: https://www.nihr.ac.uk/explore-nihr/funding-programmes/public-health-research.htm

### This report

The research reported in this issue of the journal was funded by the PHR programme as project number 15/181/01. The contractual start date was in January 2017. The final report began editorial review in September 2021 and was accepted for publication in May 2022. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The PHR editors and production house have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the final report document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health and Care Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, the PHR programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, the PHR programme or the Department of Health and Social Care.

Copyright © 2023 Lohan *et al.* This work was produced by Lohan *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This is an Open Access publication distributed under the terms of the Creative Commons Attribution CC BY 4.0 licence, which permits unrestricted use, distribution, reproduction and adaptation in any medium and for any purpose provided that it is properly attributed. See: https://creativecommons.org/licenses/by/4.0/. For attribution the title, original author(s), the publication source – NIHR Journals Library, and the DOI of the publication must

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland, and final files produced by Newgen Digitalworks Pvt Ltd, Chennai, India (www.newgen.co).

### NIHR Journals Library Editor-in-Chief

Dr Cat Chatfield Director of Health Services Research UK

### **NIHR Journals Library Editors**

**Professor Andrée Le May** Chair of NIHR Journals Library Editorial Group (HSDR, PGfAR, PHR journals) and Editorin-Chief of HSDR, PGfAR, PHR journals

**Dr Peter Davidson** Interim Chair of HTA and EME Editorial Board. Consultant Advisor, School of Healthcare Enterprise and Innovation, University of Southampton, UK

**Professor Matthias Beck** Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Consultant in Public Health, Delta Public Health Consulting Ltd, UK

Ms Tara Lamont Senior Adviser, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Dr Catriona McDaid Reader in Trials, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Emeritus Professor of Wellbeing Research, University of Winchester, UK

**Professor James Raftery** Professor of Health Technology Assessment, School of Healthcare Enterprise and Innovation, University of Southampton, UK

**Dr Rob Riemsma** Consultant Advisor, School of Healthcare Enterprise and Innovation, University of Southampton, UK

**Professor Helen Roberts** Professor of Child Health Research, Child and Adolescent Mental Health, Palliative Care and Paediatrics Unit, Population Policy and Practice Programme, UCL Great Ormond Street Institute of Child Health, London, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

**Professor Helen Snooks** Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk